
Great Basin Incident Commander Conference Call Agenda 2016 

 
Date:      Friday, August 05, 2016 Note taker:  Gina Bald 
GB MAC/GACC Coordinator:   
   
Agenda/Incident Commander Discussion Points: Summarize the following in 4 min or less: 
• Fire Size / % contained / # of people / Cost to date 
• Significant changes since the last 209 
• BRIEF fire synopsis /Critical Incident Information (use Directional vs DIVS’s) 
• Political Issues / Community Meetings 
• Closures / Evacuations / Public Safety Issues 
• Major Safety Concerns / Injuries / Accidents 
• Critical Resource Needs 
• Resources Available to Reassign/Share/Demob 
 
Predictive Services: 

Weather update: 
Shelby Law 

Next few days:  Moisture in NW of the GB.  Some precipitation in E. NV.  

Sunday:  Dry in southern GB with thunderstorms in Central ID.  Breezy 

conditions come in late weekend and early next week. Thunderstorms move 

up.  Increased Initial Attack in ID and NW UT.  Potential for large fires. 
GBCC Update: 
Gina Dingman 

• Minimal IA.  2 new starts for .2 acres 
• Large Fire Growth: 17734 
• 2 MAFFS in BOI, 5 ATs, 1 VLAT, in PIH O/S.  2 LPs (a third 

available at 1630), 2 ASMs, 4 loads of Smokejumpers at 4 bases, 
22 SEATs with 2 on D/O. 

• Reallocating resources in Staging Areas in SLC, TWF and, EKO 
 

1 
Fire: Pioneer 
IC: Beth Lund 
Report:  Tooter 
Burdick  
 

51903 / 27% /1489 /21.6M  
• West:  Heel of Fire.  All lined and holding and looking good.  

Started burning out.  Hope to complete today. Lowman:  line in to 
prescribed burn – tied off.  East side of Rock Creek is burnt. 

• East:  Old Lowman burn.  Some spots.  Head of fire didn’t move 
much.  Hit Rabbit fire – not moving fast.  Grasses drying out 
rapidly. 

• Critical Resource Needs:  2 T2 IHC 
• Evacuations:  Still Level 1 status in Lowman 
• Political issues:  Hwy 21 still closed between Idaho City and 

Lowman.  Meeting today with IDT and Forest to get hwy back 
open. 

• Injuries:  10 Total, 2 with lost time 
• Resources available for release: None 

2 
Fire:  Broad Mouth 

19000 est (No GIS) / 30% / 29% / $750000 



IC:   
Report:  Snider 

• South:  Heel of fire 90% lined and anchored indirect.  Tough 
country for dozers.  Firing operations today. 

• West:  80% indirect dozer line.  Firing operations today. 
• North:  Open line. Some indirect dozer line. Major or activity.  

Dozer line in today. 
• Injuries:  None 
• Some media attention 
• Sage Grouse habitat, smoke in the SLC valley 
• No Critical Resource Needs 
• Resources available for Release:  None 

3 
Fire:  Sheep Creek 
IC:  Armantrout 
Report:  Brook 
Chadwick 

1061 / 60% /160  /$725000 
• Secure and mop-up interior 
• No political issues 
• Closures:  Lifted 
• Critical Resource Needs:  None 
• Resource to assign:  2 T2 Crews, Boise IHC end of shift 

4 
Fire: Cliff Creek 
IC: Russ Bird 
Report:  Kuo 
 
 
 
 

31662/ 84% / 336 /12.6M 
• Changes since last 209 submitted:  Slight increase in acres  
• Calm, everything looking good 
• West:  Mop-up, suppression repair.  Allowing re-entry and 

energizing powerline. 
• South and North:  Patrol by air.  Natural barrier. 
• East:  Securing 
• Political Issues:  Meeting in Hoback.  Re-entry on west side at 1000.  

Lifted evacuations.   
• Closures:  North of Hwy.  Limited public access to those unaffected 

by fire 
• Injury/Accident:  None    
• Critical Resource Needs:  None    
• Resources to reassign: None.  Glidepath shows resources staying 

with WY T3 Team (IC Swenson).  Transfer of command on Sunday. 
5 
Virginia Mts Complex 
IC: Tim Roide 

59727 / 73% /607  /7.3M   
• No significant changes to 209 Report 
• Tule and Anderson: Secure and mop up, suppression repair 
• All resources off end of shift today   
• No Closures  
• No injuries 
• Critical Resource Needs:  None 
• Meeting with Paiute Tribe 
• Resources available:  All remaining operational resources 

available today and Saturday 
• Closeout today at 1500 

 
6 
Fire: Box Canyon 
IC: Pilgrim 

2203 / 0% / 76 / $498000 
• Changes since last 209 reported:  235 acreage gain 



• 70% of incident is in monitor and patrol status; 30 % taking 
action. (0% contained) 

• Weather calm.  Addition on IMET helping with more accurate 
forecasts. 

• 40 acre burn  on top of Hell’s Kitchen basin 
• Long-term assessment of border communities 
• Public meeting:  Oakley last night.  60 local residents. Pleased 

with progress. 
• Closures:  Campground North of Fire.  Closure postponed until 

next week.  Hiking trails and 4x4 Roads closed. 
• No injuries 
• Critical Resource Needs: Long-term T3 Helo to relieve local IA ship    
• Resources to assign:  1 T2 Crew with 10 days left 

7 
Fire:  Rocky Peak 
IC:  Berkibile 
Report:  Simonson 

950 / 80% / 60 / $70000 
• East of Pocatello – IA from Wednesday 
• Critical Resource Needs:  None 
• No Injuries 
• Most resources released this afternoon 

5 
Poodle 
IC:  Uhlig 
 

6557 / 95% /237 /1.5 
• No Evacuations 
• No Injuries 
• No Critical Resource Needs 
• 7 T2 Crews leaving ICP today 

8 
Fire: Comet 
IC: McPhetridge 
Report:  Terry 
Swinscoe 

358 / 90% / 9 / 1.3M 
• No injuries 
• Critical Resource Needs: None 
•  

MAC / GBCC 
Coordinator 
Derrek Hartman 

With last few days on minimal IA and large fire activity, looking at Mac 
Glidepath.  Looking at 2-3 day events for T1 Crews – 24-32 hour perspective.  
Thanks for your support. 

Date/Time of Next Call:   8/6 @ 0800 MT / 0900 PT  
 


